Author's response to reviews

Title: Optimising Human Resources in Health Care: From a Staff-Mix Perspective Towards a Systemic Approach

Authors:

Carl-Ardy Dubois (carl.ardy.dubois@umontreal.ca)
Debbie Singh (debbie.singh@arachna.co.nz)

Version: 3 Date: 29 October 2009

Author's response to reviews:

1. Correct the doc you uploaded for fig 1 (it currently shows a blank page): DONE
2. Please put both authors' full addresses: DONE
3. Correct reference numbering style throughout the doc: we do not accept superscript - numbers should be in square brackets, before any punctuation marks. e.g., [1]. DONE
4. Similarly, put normal numbers (not superscript) at the end of the doc in the reference section for each reference. e.g., 1. DONE